Author's response to reviews

Title: Can Hyperbaric Oxygen be Used to Prevent Deep Infections in Neuro-muscular Scoliosis Surgery?

Authors:

Mustafa E İnanmaz (erkaninanmaz@yahoo.com)
Kamil C Kose (kacako@hotmail.com)
Cengiz Isik (drcengiz034@yahoo.com)
Halil Atmaca (drhalilatmaca@hotmail.com)
Hakan Basar (hbasar80@hotmail.com)

Version: 2 Date: 17 September 2013

Author's response to reviews:

Dear Sir,

Below are the changes that have been made during revision of our manuscript according to your suggestions.

1.) Title page is now included in the main manuscript.
2.) A "Competing interests" section was added below the references section.
3.) An "Authors' contributions" section was added below the references section.
4.) An “Acknowledgements section was added below the references section.
5.) A reference to the ethics statement was made below the references section
6.) A reference to the written permission obtained from the patient for publication of figure 1 was made below the references section.

All changes were highlighted in yellow.

Sincerely
Mustafa Erkan Inanmaz, M.D.